Geophysical Research Letters

RESEARCH LETTER

10.1029/2019GL084303

Key Points:

- Permafrost was incubated in seawater to determine the amount of CO₂ potentially released by eroding permafrost coasts
- Observed CO_2 production after 4 months is up to 4.3 ± 1.0 mg/gdw at 4 °C and comparable to release rates from thawing permafrost on land
- Permafrost coasts and nearshore waters are potentially an important source of CO₂ that is neglected in carbon cycle budgets and models

Supporting Information:

- Supporting Information S1
- Figure S1
- Figure S2
- Figure S3
- Figure S4
- Figure S5Figure S6
- Figure S6Data Set S1
- Data Set S1
 Data Set S2

Correspondence to:

G. Tanski, george.tanski@vu.nl

Citation:

Tanski, G., Wagner, D., Knoblauch, C., Fritz, M., Sachs, T., & Lantuit, H. (2019). Rapid CO₂ release from eroding permafrost in seawater. *Geophysical Research Letters*, 46. https://doi.org/ 10.1029/2019GL084303

Received 4 JUL 2019 Accepted 12 SEP 2019 Accepted article online 15 OCT 2019

©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Rapid CO₂ Release From Eroding Permafrost in Seawater

G. Tanski^{1,2}, D. Wagner^{3,4}, C. Knoblauch^{5,6}, M. Fritz², T. Sachs³, and H. Lantuit^{2,4}

¹Faculty of Science, Earth and Climate, Vrije Universiteit Amsterdam, Amsterdam, The Netherlands, ²Permafrost Research Unit, Alfred Wegener Institute, Helmholtz Centre for Polar and Marine Research, Potsdam, Germany, ³GFZ German Research Centre for Geosciences, Potsdam, Germany, ⁴Institute of Geosciences, Potsdam University, Potsdam, Germany, ⁵Institute of Soil Science, Universität Hamburg, Hamburg, Germany, ⁶Center for Earth System Research and Sustainability, Universität Hamburg, Hamburg, Germany

Abstract Permafrost is thawing extensively due to climate warming. When permafrost thaws, previously frozen organic carbon (OC) is converted into carbon dioxide (CO₂) or methane, leading to further warming. This process is included in models as gradual deepening of the seasonal non-frozen layer. Yet, models neglect abrupt OC mobilization along rapidly eroding Arctic coastlines. We mimicked erosion in an experiment by incubating permafrost with seawater for an average Arctic open-water season. We found that CO₂ production from permafrost OC is as efficient in seawater as without. For each gram (dry weight) of eroding permafrost, up to $4.3 \pm 1.0 \text{ mg CO}_2$ will be released and $6.2 \pm 1.2\%$ of initial OC mineralized at 4 °C. Our results indicate that potentially large amounts of CO₂ are produced along eroding permafrost coastlines, onshore and within nearshore waters. We conclude that coastal erosion could play an important role in carbon cycling and the climate system.

Plain Language Summary The permanently frozen soils of the Arctic, known as permafrost, store large amounts of organic carbon, which accumulated over millennia due to slow decomposition in the cold Arctic regions. With climate warming this frozen organic carbon reservoir thaws and microbes recycle it quickly into greenhouse gases, which in turn support further warming. A slow and continuous thaw is currently used in models to project future greenhouse gas release from permafrost. Yet along the rapidly eroding coastlines of the Arctic Ocean, which make up 34% of the Earth's coastlines, whole stretches of the coast simply collapse, sink or slide into the ocean, including the previously frozen organic carbon. We simulated greenhouse gas release in response to coastline collapse in a laboratory experiment by simply mixing permafrost with seawater. We show that large amounts of carbon dioxide are being produced during the Arctic open-water season. Our study indicates that eroding permafrost coasts in the Arctic are potentially a major source of carbon dioxide. With increasing loss of sea ice, longer open-water seasons, and exposure of coasts to waves, we highlight the importance of coastal erosion for potential carbon dioxide emissions.

1. Introduction

Soils in the permafrost region of the Northern Hemisphere store an estimated 1,300 Pg $(1.3 \times 10^{18} \text{ g})$ of organic carbon (OC; Hugelius et al., 2014), an amount that exceeds the pool of carbon in the atmosphere (Ciais et al., 2013). With global warming, higher temperatures are causing permafrost to thaw and its OC to become subject to microbial turnover, producing greenhouse gases (GHGs; Knoblauch et al., 2013; Schädel et al., 2014), which further fuel global warming (Koven et al., 2015; Schuur et al., 2015). This process is represented in models as a one-dimensional (vertical) seasonal deepening of the active (non-frozen) layer. Although studies highlight the importance of abruptly thawing permafrost (Turetsky et al., 2019) and first attempts were made to include thermokarst (i.e., ground disturbance and subsidence caused by thawing of ice-rich permafrost; Schuur et al., 2015) and erosion processes on land in climate models (Lee et al., 2014; Von Deimling et al., 2015), this representation neglects the rapid mobilization of permafrost OC due to coastal erosion and thermokarst and thus OC turnover during lateral transport from land to sea (Figure 1). This process is likely to substantially increase GHG emissions from thawing permafrost (Fritz et al., 2017; Vonk & Gustafsson, 2013).

Permafrost coasts make up approximately 34% of the Earth's total coastline (Lantuit et al., 2012). Erosion of these coasts releases large amounts of permafrost OC directly into the Arctic Ocean (Couture et al., 2018;







Figure 1. Lateral erosion of permafrost coasts in the Arctic. Coastal erosion rapidly mobilizes organic carbon pools from permafrost that can be quickly turned over into carbon dioxide onshore and within the nearshore zone (a). Figure 1a illustrates the lateral carbon flux from coastal erosion identified in this study (yellow vertical arrow) that is neglected in current models. Figures 1b and 1c give examples of erosion modes in form of mud lobes (b) and block failures (c) and conceive the experimental approach with permafrost (PF) and permafrost mixed with seawater (PF + SW) mimicking coastal erosion under laboratory conditions. Carbon dioxide (CO₂) and methane (CH₄) were measured during incubations of PF and PF + SW for the length of an average Arctic open-water season of 4 months. Pictures were taken on *Qikiqtaruk*—Herschel Island (Yukon Territory, Canada).

Wegner et al., 2015). Thermokarst and erosional features on land, which can occupy up to 20% of the permafrost landscape (Olefeldt et al., 2016), can further contribute to this OC flux by mobilizing deep permafrost carbon pools before waves erode the shore (see Figure 1). The most prominent features are retrogressive thaw slumps, thermoerosional gullies, and active layer detachment slides (Abbott et al., 2015) and are initiated by thaw of massive ice beds or degradation of ice wedges (Jorgenson et al., 2015; Ramage et al., 2017). The occurrence of thermoerosional features increases the amount of OC transported to the ocean but also leads to OC degradation prior to the release into the ocean (Tanski et al., 2017; Vonk et al., 2013; Ward & Cory, 2016). In short, the mechanisms of erosion at the Arctic coast are multifaceted but all contribute to the release of OC to the nearshore zone. The current mean circum-Arctic erosion rate is 0.57 m/year but can exceed 20 m/year at specific locations (Jones et al., 2018), inducing the transfer of large masses of sediment (430 Tg/year) and OC (up to 14 Tg/year; Wegner et al., 2015) to the nearshore



Figure 2. Exemplified sampling profile of exposed permafrost sediments. Sampling targeted the upper organic-enriched permafrost layer (PF_{ORG}), the deeper mineral permafrost layer (PF_{MIN}), and seawater ~1 km off the coast of the sampling location. Depth is given in meter below surface. For original sampling profiles and sampling locations see Figures S1 and S2.

zones of the Arctic Ocean. This carbon is potentially subject to mineralization into GHGs within the coastal realm upon release from permafrost (Semiletov et al., 2013; Vonk et al., 2012; Vonk & Gustafsson, 2013). Projected higher water temperatures and longer periods of open-water (ice-free) conditions will further promote the erosion of these coasts as cliffs will be exposed to wave action and storms during longer time spans (Barnhart et al., 2014; Dai et al., 2019; Overeem et al., 2011). Several studies already show that permafrost OC turnover in aquatic systems is substantial and important (Abbott et al., 2014; Vonk et al., 2012; Vonk et al., 2015). Yet in recent carbon budgets, OC from coastal erosion is assumed to be subject to turnover only at the cliff (Vonk et al., 2012) with the remainder exported offshore and buried in seafloor sediments (McGuire et al., 2009). Here, we hypothesize that a substantial part of this OC is not buried or exported but directly released from the coastal zones, onshore and within nearshore waters, as carbon dioxide (CO₂) in magnitudes relevant to the Arctic carbon budget and climate system.

2. Materials and Methods

2.1. Mimicking Coastal Erosion and GHG Production With Incubations

Insights on GHG emissions from eroding permafrost coasts in the Arctic are extremely rare. We developed a mechanistic approach to simulate erosion under laboratory conditions by mixing permafrost material with seawater and quantifying the subsequent GHG release. Sampling of permafrost (PF) and seawater (SW) took place on *Qikiqtaruk*—Herschel Island at the Yukon coast (Figure S1 in the supporting information). We targeted the upper (<1 m depth) organic-enriched layer (PF_{ORG;} here defined as permafrost with a total OC (TOC) content of >5.0 wt%) and the lower (>1 m depth) mineral permafrost layer (PF_{MIN}; Figures 2 and S2). The average TOC content of our samples (4.7 \pm 3.1 wt%) is slightly lower than carbon contents observed in continuous permafrost zones of

Alaska and Siberia ($10.5 \pm 13.8\%$ TOC; Schädel et al., 2014, and references therein) but is in the range of average TOC contents calculated for the entire Beaufort Sea and circum-Arctic coastline with 2.4–5.7 and 2.1 wt% OC, respectively (Lantuit et al., 2012). The seawater used for incubation was taken 1 km offshore the sampling site, where turbidity due to erosion or resuspension was not occurring. The Yukon Coast is strongly affected by erosion and thermokarst formation (Irrgang et al., 2019; Ramage et al., 2017; Figure 1). During a brief open-water period of approximately 4 months, coastal cliffs on *Qikiqtaruk*— Herschel Island erode at an average rate of 0.45 m/year (1970–2000; Lantuit & Pollard, 2005) and retrogressive thaw slumps are highly active (Ramage et al., 2017). The local erosion rate is slightly lower than for the adjacent Yukon Coast (0.7 m/year; Couture et al., 2018) and Beaufort Sea coast (1.1–1.2 m/year) but close to the circum-Arctic rate of 0.57 m/year (Lantuit et al., 2012).

Qikiqtaruk—Herschel Island is a moraine formed during the late Wisconsin glaciation and characterized by continuous permafrost (Pollard, 1990). Permafrost is covered by a seasonally non-frozen active layer varying between 40 and 60 cm in thickness during the summer season (Burn & Zhang, 2009). Sediment composition and ground ice contents (45–75 vol.%; Couture & Pollard, 2015) are similar to the rest of the western Canadian Arctic, Alaska, and most of Siberia. In these regions sediments are mainly composed of unconsolidated fine-grained deposits with high ground ice contents of 50–60 vol.% in the western Canadian Arctic (French, 1998), 44–89 vol.% in Alaska (Kanevskiy et al., 2013) and 65–90 vol.% in Siberia (Schirrmeister et al., 2011).

2.2. Experimental Setup and Incubation Conditions

To mimic GHG production during coastal erosion and in situ conditions at the coast, three aliquots of each PF_{ORG} (n = 2) and PF_{MIN} (n = 2) sample were incubated with seawater for an average Arctic open-water



Table 1

Summary of Incubation Setups, Cumulative CO_2 Production, and Associated TOC Loss

Т	Sample	n	CO ₂ -C (mg/ gdw)	std (±)	TOC loss (%)	std (±)
(4 °C)	PF _{ORG} + SW	6	4.3	1.0	1.5	0.2
	PF _{ORG} Control	2	3.1	0.3	1.1	0.0
	$PF_{MIN} + SW$	6	3.6	0.6	6.2	1.2
	PF _{MIN} Control	2	2.3	0.3	3.9	0.2
(16 °C)	$PF_{ORG} + SW$	6	5.5	0.7	1.9	0.1
	PF _{ORG} Control	2	6.4	1.4	2.4	0.3
	$PF_{MIN} + SW$	6	7.4	1.1	12.8	1.9
	PF _{MIN} Control	2	3.9	0.3	6.8	0.4

Note. Coastal erosion was mimicked by mixing representative permafrost deposits (see Figure 2) with seawater (SW) at different temperatures (*T*) under aerobic conditions for an average Arctic open-water season of 4 months, that is, 120 days. Organic-enriched and mineral permafrost (PF_{ORG} and PF_{MIN} , respectively) were incubated with SW and separately for control conditions (see supporting information Table S1). Total organic carbon (TOC) loss in percent is given as loss in form of CO₂ from initial TOC (mg C/gdw) measured prior to the incubations in PF sediments.

season of 4 months, that is, 120 days (Lantuit et al., 2012) (Table 1; see Figures S2, S3, and Text S1). Aerobic conditions were applied to mimic in situ conditions at the coast. Low and high temperatures (4 and 16 °C, respectively) were applied to study the effect of increased temperatures on OC decomposition. GHG emissions in form of CO₂ and methane (CH₄) were measured during the incubations (see Text S2) and complemented with analyses of TOC, dissolved organic carbon (DOC), stable carbon isotopes (δ^{13} C-TOC), and TOC/total nitrogen (TOC/TN) ratios (see Text S3) and measured before and after 4 months of incubation (T = 1) to determine carbon turnover (Knoblauch et al., 2013; Strauss et al., 2015). OC loss and turnover into CO2 was estimated using initial TOC content and produced CO_2 (mg C/gdw) at the end of the incubation (see Text S4). We used the 4 °C temperature as reference for local temperatures in the study area, since it is in the range of temperatures observed along the Yukon Coast in summer and comparable to other incubation studies (Burn & Zhang, 2009; Lee et al., 2014; Schädel et al., 2014). We used the 16 °C scenario as the plume of the Mackenzie River can transport warm water masses to the

Yukon Coast (Dunton et al., 2006) and to be comparable with previous studies that used a 15 °C temperature setting (e.g., Dutta et al., 2006; Lee et al., 2012).

3. Results and Discussion

3.1. CO₂ Production and Carbon Turnover

The experiment shows that substantial amounts of CO_2 are produced from thawing permafrost mixed with seawater and without, indicating CO_2 release along the land-ocean continuum of eroding permafrost coastlines. During an average open-water season of 4 months, between 3.6 ± 0.6 (PF_{MIN} + SW) and 4.3 ± 1.0



Figure 3. Cumulative aerobic CO_2 production from coastal permafrost during an incubation period of 120 days. Samples were incubated at 4 °C (a, b) and 16 °C (c, d). Cumulative aerobic CO_2 production is given ± standard deviation (error bars) for organic-enriched permafrost (PF_{ORG}; a, c) and mineral permafrost (PF_{MIN}; b, d) mixed with seawater (SW) and control conditions.

 $(PF_{ORG} + SW)$ mg CO₂-C/gdw are produced from thawing permafrost in seawater at 4 °C (Table 1 and Figure 3). The produced CO₂ from permafrost thaw in seawater outnumbered production from controls mimicking erosion onshore. In PF_{ORG} and PF_{MIN} Controls CO₂ production was 3.1 ± 0.3 and 2.3 ± 0.2 CO₂-C/gdw, respectively. In all incubation scenarios substantial amounts of initial OC were mineralized. The observed CO₂ production in PF_{ORG} and PF_{MIN} mixed with seawater corresponds to TOC (mg C/gdw) losses of 1.5 ± 0.2% and 6.2 ± 1.2%, respectively, whereas in PF_{ORG} and PF_{MIN} Controls, 1.1 ± 0.0% and 3.9 ± 0.2% TOC were lost, indicating quick mineralization of OC into CO₂ in the nearshore zone. The measured CO₂ release is in the range of or even greater than rates found in Alaska and Siberia, where productions of 1.07 to 3.42 mg CO₂-C/gdw (over 500 days; Lee et al., 2012) and 0.87 ± 0.56 to 1.6 ± 1.2 mg CO₂-C/gdw (over 1,200 days; Knoblauch et al., 2013) were observed for much longer incubation periods in aerated thawing terrestrial permafrost.

At higher temperatures (16 °C), total CO₂ production further increased by 21% (PF_{ORG} + SW) and 52% (PF_{MIN} + SW) when comparing the cumulated production at the end of the incubation (supporting information Figure S5). Only small amounts of methane were released (<10 µg CH₄-C/gdw; see supporting information Figure S6 and Table S1), potentially by outgassing from soil pore spaces or microbial production (Wagner et al., 1999). Organic matter decomposition is generally indicated by the decrease (given in percent change from initial measurements) of (1) DOC concentrations (28.0% in PF_{ORG} and 52.2% in PF_{MIN}), (2) TOC/TN ratios (12.8% in PF_{ORG} and 18.3% in PF_{MIN}), and an increase of δ^{13} C-TOC (0.9% in PF_{ORG} and 0.7% in PF_{MIN}) measured before and after incubations under control conditions at 4 °C (see Tables S2 and S3), which are all strong predictors for degradation and CO₂ production from permafrost OC (Knoblauch et al., 2013; Schädel et al., 2014; Strauss et al., 2015).

3.2. CO₂ Production Dynamics

The bulk of aerobic CO₂ was released quickly during the first 2 months of incubation most likely due to rapid removal of the most labile OC fractions of the fast carbon pool (Schädel et al., 2014), resulting in high CO₂ production rates over the first weeks of incubation (Dutta et al., 2006; Knoblauch et al., 2013). Maximum daily production rates peaked at 244.5 \pm 57.4 µg CO₂-C/gdw/day for PF_{ORG} + SW and 188.4 \pm 187.0 µg CO₂-C/gdw/day for PF_{MIN} after 11 days. Daily production rates observed at 16 °C were substantially higher than at 4 °C (supporting information Figure S5). Production rates are in good agreement with existing permafrost incubation studies (Knoblauch et al., 2013; Lee et al., 2012; Schädel et al., 2014), where higher OC contents result in higher CO₂ production and production rates decrease toward baseline rates after 1 to 3 months (Knoblauch et al., 2013; Lee et al., 2012; Schädel et al., 2014).

The removal of OC during the course of our experiment was particularly efficient for the labile DOC fraction, typical for permafrost OC (Abbott et al., 2014; Vonk et al., 2013) shown for the Controls (see Table S2). The presence of saltwater potentially supported the mobilization of labile OC compounds from permafrost sediments as flocculation processes in seawater discriminate against less labile high molecular weight fractions in sediments (Dou et al., 2005). This might explain higher CO_2 production in permafrost incubated with seawater than without. Aerobic microbial communities with very high respiration rates found in aerated top layers of shallow coastal sediments are probably responsible for the quick CO_2 production in seawater (Arndt et al., 2013; Rasmussen & Jorgensen, 1992). The time required to completely convert the OC pool exceeds the duration of a single open-water period (Knoblauch et al., 2013) and was not accounted for in the experiment. Yet it implies that OC decomposition and CO_2 production could continue on longer time-scales if not buried and preserved on the seafloor (Bröder et al., 2018; Schädel et al., 2014; Vonk et al., 2013).

Our results question the paradigm in current carbon budgets that OC is entirely transported offshore, utilized for primary production or buried in shelf sediments (Bröder et al., 2016; Dunton et al., 2006; Vonk & Gustafsson, 2013). We suggest an additional mechanism of GHG production from coastal permafrost degradation and collapse into the Arctic nearshore zone. We show that a fraction of eroded permafrost OC is quickly mineralized into CO_2 , with or without seawater added, indicating potential CO_2 release to the atmosphere during coastal erosion. This corresponds with findings from Siberia that showed extensive outgassing of CO_2 from the coastal zone and oversaturated CO_2 conditions in coastal waters, potentially as a result of OC mineralization from terrestrial sources (Semiletov et al., 2013) or studies from the Yukon Coast, which show that only ~13% of eroded OC is sequestered in nearshore sediments with the remainder being potentially available for turnover processes (Couture et al., 2018).

3.3. OC Pathways and CO₂ Dynamics in the Coastal Zone

Latest carbon budgets identify the coastal zone and the continental shelves of the Arctic Ocean as a CO₂ sink (Bates & Mathis, 2009; McGuire et al., 2009). However, we show that coastal permafrost thawing in seawater produces large amounts of CO₂ over a short period. Thus, eroding permafrost coasts and its nearshore waters may release substantial amounts of CO_2 during the open-water season onshore and within the nearshore waters. This carbon pathway and the associated GHG release is not accounted for in climate models yet and might have been already an important contributor during postglacial sea level rise and erosion of the former Beringian landmass (Wegner et al., 2015). To illustrate the potential significance of our findings at a circum-Arctic scale, we used a simple approach to estimate potential CO₂ emissions from coastal collapse on an annual basis. We projected the minimum and maximum observed TOC loss at 4 °C (1.1 ± 0.0 and $6.2 \pm 1.2\%$ over 120 days at 4 °C; Table 1) derived from our experiment to the most recent estimation of TOC fluxes from Arctic coasts (up to 14.0 Tg/year; Wegner et al., 2015). This calculation yields an annual CO₂ release rate between 0.2 ± 0.0 to 0.9 ± 0.2 Tg/year from permafrost erosion on land and in nearshore waters. The TOC loss of our study generally corresponds to findings from Dutta et al. (2006), who incubated permafrost over 90 days at 5 °C and showed an OC loss as GHG of $2.1 \pm 0.2\%$. Our calculated CO₂ release compares with the 0.5-1 Pg OC/year projected to be released from vertical permafrost thaw (Schuur et al., 2008) and is in the same order of magnitude as other carbon fluxes in the permafrost environment such as OC fluxes from Arctic rivers into the ocean with 5.4 Tg POC/year and 25 Tg DOC/year (Macdonald et al., 2015; Raymond et al., 2007) or methane fluxes from northern and thermokarst lakes with 24.2 \pm 10.5 Tg CH₄/year (Walter et al., 2007). Although our estimated number is only conservative, the calculated CO_2 release from coastal erosion onshore and within the seawater column is substantial and has not been considered for carbon budgets and climate simulations so far.

Our observed CO_2 production and calculated release from nearshore waters comes with several uncertainties. In the laboratory we had to neglect environmental parameters, which could not be mimicked but that most likely have a strong impact on CO_2 pathways in the coastal environment. This includes the environmental variability of OC stocks along Arctic coasts and its vulnerability to thaw and microbial turnover into GHGs, length of the open-water season, presence and duration of phytoplankton blooms, pH, water temperature, and oxygen availability as well as the depositional dynamics, bathymetry, water depth, and nearshore currents along Arctic coasts. Yet we believe that terrestrial OC turnover and potential CO_2 production is promoted along the coastal margins due to the combined effects of resuspension and supply of oxygen in shallow nearshore waters (Fritz et al., 2017; Macdonald et al., 2015; Vonk et al., 2012) or in lagoon systems and submerged areas of the Arctic (Brown et al., 2003; Dunton et al., 2006; Ruz et al., 1992; Schreiner et al., 2013). From the uncertainties mentioned, especially ice algae and primary production stimulated by nutrient supply from upwelling or terrestrial input (Ardyna et al., 2014; Tremblay et al., 2011) could promote the uptake of produced CO_2 stemming from permafrost erosion in coastal waters but turbidity caused by erosion and resuspension potentially may limit this uptake (Semiletov et al., 2013).

Terrestrial OC is being buried in large amounts directly in nearshore sediments (Couture et al., 2018; Vonk et al., 2014). Yet not the entire eroded OC is sequestrated in nearshore sediments and substantial portions are transported within the water column (Vonk et al., 2014). Stein and Macdonald (2004) estimated that ~33% of terrestrial-derived particulate OC is metabolized in the ocean. Heavier portions of mineral-bound terrestrial OC and older OC carbon fractions might quickly settle and accumulate upon release into coastal waters (Gustafsson et al., 2000; Karlsson et al., 2011; Vonk et al., 2014). Yet waves, longshore currents, and ice scouring can keep this OC in resuspension (Are, 1988; Macdonald et al., 2015; Vonk et al., 2012) making it available for further degradation into DOC and thus microbial uptake and decomposition into CO_2 (Battin et al., 2008; Dou et al., 2005). OC stemming from permafrost can contribute between <10% and almost 50% of OC in nearshore waters of the Arctic Ocean (Vonk et al., 2014). Labile terrestrial DOC fractions can be quickly mineralized within the water column and have been shown to have a short residence time of approximately 2–5 years on the shelf (Alling et al., 2010; Letscher et al., 2011) during which approximately 30-50% of DOC is potentially being lost. Sunlight may further catalyze the turnover process in the upper water column by photooxidation (Cory et al., 2014; Ward et al., 2017). Eventually, resuspension and bioturbation also remobilize buried OC and slow carbon pools (Arndt et al., 2013; Rasmussen & Jorgensen, 1992), which decomposition can take up to decades (Schädel et al., 2014), postponing CO₂ production beyond one openwater season.

4. Conclusion

We show that substantial amounts of CO_2 are released from thawing permafrost in seawater during incubations. We conclude that potentially large amounts of CO_2 are being produced along eroding permafrost coasts, onshore during transit and within the nearshore water column, especially in the shallow water column where resuspension occurs. This carbon and GHG release pathway is not accounted for yet in climate models. We expect that coastal CO_2 production further increases due to the combined effects of accelerated coastal erosion, higher sea surface temperatures, and a longer open-water season which increases the time when cliffs are exposed to wave action and storm events. The high pace of environmental change along Arctic permafrost coasts could catalyze and increase the production of CO_2 in the coastal realm and thus strengthen permafrost carbon cycling.

Acknowledgments

We are grateful for the financial support of the Helmholtz Young Investigators Groups "COPER" (Grant VH-NG-801 to H. Lantuit) and "TEAM" (Grant VH-NG-821 to T. Sachs), the Daimler and Benz Foundation (Grant 32-02/15 to M. Fritz), and Nunataryuk (Grant 773421). We thank O. Burckhardt and I. Milczarek for their support in the lab and J. Strauss, G. Schwamborn, S. Liebner, A. Kiss, B. Koch, and J. Vonk for their expertise and scientific guidance, A. M. Irrgang, E. McLeod, and S. Stettner are thanked for assisting in the field and J. Kahl for logistical support. We especially acknowledge the logistical support in Canada provided by the Aurora Research Institute Inuvik and R. Gordon and the rangers from the Qikiqtaruk-Herschel Island Territorial Park. Finally, we thank C. O'Connor for language correction and proofreading as well as the Editor and reviewer for their helpful comments. Additional and supporting data can be accessed in the PANGAEA data repository (https://doi. pangaea.de/10.1594/ PANGAEA.905912).

References

- Abbott, B. W., Jones, J. B., Godsey, S. E., Larouche, J. R., & Bowden, W. B. (2015). Patterns and persistence of hydrologic carbon and nutrient export from collapsing upland permafrost. *Biogeosciences*, *12*(12), 3725–3740. https://doi.org/10.5194/bg-12-3725-2015
- Abbott, B. W., Larouche, J. R., Jones, J. B., Bowden, W. B., & Balser, A. W. (2014). Elevated dissolved organic carbon biodegradability from thawing and collapsing permafrost. *Journal of Geophysical Research: Biogeosciences*, 119, 2049–2063. https://doi.org/10.1002/ 2014JG002678
- Alling, V., Sanchez-Garcia, L., Porcelli, D., Pugach, S., Vonk, J. E., Dongen, B. E., et al. (2010). Nonconservative behavior of dissolved organic carbon across the Laptev and East Siberian seas. *Global Biogeochemical Cycles*, 24, GB4033. https://doi.org/10.1029/ 2010GB003834
- Ardyna, M., Babin, M., Gosselin, M., Devred, E., Rainville, L. A., & Tremblay, J. E. (2014). Fall phytoplankton blooms. Geophysical Research Letters, 41, 6207–6212. https://doi.org/10.1002/2014GL061047.Received
- Are, F. E. (1988). Thermal abrasion of sea coasts (part I). Polar Geography and Geology, 12(1), 1. https://doi.org/10.1080/ 10889378809377343
- Arndt, S., Jørgensen, B. B., LaRowe, D. E., Middelburg, J. J., Pancost, R. D., & Regnier, P. (2013). Quantifying the degradation of organic matter in marine sediments: A review and synthesis. *Earth-Science Reviews*, 123, 53–86. https://doi.org/10.1016/j.earscirev.2013.02.008
- Barnhart, K. R., Anderson, R. S., Overeem, I., Wobus, C., Clow, G. D., & Urban, F. E. (2014). Modeling erosion of ice-rich permafrost bluffs along the Alaskan Beaufort Sea coast. *Journal of Geophysical Research: Earth Surface*, 119, 1155–1179. https://doi.org/10.1002/ 2013JF002845
- Bates, N. R., & Mathis, J. T. (2009). The Arctic Ocean marine carbon cycle: Evaluation of air-sea CO₂ exchanges, ocean acidification impacts and potential feedbacks. *Biogeosciences*, 6(4), 2433–2459. https://doi.org/10.5194/bgd-6-6695-2009
- Battin, T. J., Kaplan, L. A., Findlay, S., Hopkinson, C. S., Marti, E., & Packman, A. I. (2008). Biophysical controls on organic carbon fluxes in fluvial networks. *Nature Geoscience*, 1(2), 95–100. https://doi.org/10.1038/ngeo101
- Bröder, L., Tesi, T., Andersson, A., Semiletov, I., & Gustafsson, Ö. (2018). Bounding cross-shelf transport time and degradation in Siberian-Arctic land-ocean carbon transfer. *Nature Communications*, 9(1), 806. https://doi.org/10.1038/s41467-018-03192-1
- Bröder, L., Tesi, T., Andersson, A., Eglinton, T. I., Semiletov, I. P., Dudarev, O. V., et al. (2016). Historical records of organic matter supply and degradation status in the East Siberian Sea. Organic Geochemistry, 91, 16–30. https://doi.org/10.1016/j.orggeochem.2015.10.008
- Brown, J., Jorgenson, M., Smith, O., & Lee, W. (2003). Long-term rates of coastal erosion and carbon input, Elson Lagoon, Barrow, Alaska. *Eighth International Conference on Permafrost*, 21–25. Retrieved from http://www.arlis.org/docs/vol1/ICOP/55700698/Pdf/Chapter_019.pdf Burn, C. R., & Zhang, Y. (2009). Permafrost and climate change at Herschel Island (Qikiqtaruq), Yukon Territory, Canada. *Journal of Geophysical Research*, 114, F02001. https://doi.org/10.1029/2008JF001087
- Ciais, P., Sabine, C., Bala, G., Bopp, L., Brovkin, V., Canadell, J., et al. (2013). Carbon and other biogeochemical cycles. In T. F. Stocker, D. Qin, G. K. Plattner, M. Tignor, S. K. Allen, J. Boschung, et al. (Eds.), Climate change 2013: The physical science basis. Contribution of working group 1 to the fifth assessment report of the Intergovernmental Panel on Climate Change (Chap. 6, pp. 465–570). Cambridge (UK), New York (USA): Cambridge University Press.
- Cory, R. M., Ward, C. P., Crump, B. C., & Kling, G. W. (2014). Sunlight controls water column processing of carbon in arctic fresh waters. *Science*, 345(6199), 925–928. https://doi.org/10.1126/science.1253119
- Couture, N., & Pollard, W. H. (2015). Ground ice determinations along the Yukon coast using a morphological model. In Proceedings, GeoQuebec 2015, 7th Canadian Permafrost Conference. Québec City, Canada.
- Couture, N. J., Irrgang, A., Pollard, W., Lantuit, H., & Fritz, M. (2018). Coastal erosion of permafrost soils along the Yukon coastal plain and fluxes of organic carbon to the Canadian Beaufort Sea. *Journal of Geophysical Research: Biogeosciences*, 123(2), 406–422. https://doi.org/ 10.1002/2017JG004166
- Dai, A., Luo, D., Song, M., & Liu, J. (2019). Arctic amplification is caused by sea-ice loss under increasing CO₂. Nature Communications, 10(1), 1–13. https://doi.org/10.1038/s41467-018-07954-9
- Dou, F., Ping, C.-L., Guo, L., & Jorgenson, T. (2005). Estimating the impact of seawater on the production of soil water-extractable organic carbon during coastal erosion. Journal of Environmental Quality, 37(6), 2368–2374. https://doi.org/10.2134/jeq2007.0403

- Dunton, K. H., Weingartner, T., & Carmack, E. C. (2006). The nearshore western Beaufort Sea ecosystem: Circulation and importance of terrestrial carbon in arctic coastal food webs. *Progress in Oceanography*, 71(2–4), 362–378. https://doi.org/10.1016/j.pocean.2006.09.011 Dutta, K., Schuur, E. A. G., Neff, J. C., & Zimov, S. A. (2006). Potential carbon release from permafrost soils of Northeastern Siberia. *Global Change Biology*, 12(12), 2336–2351. https://doi.org/10.1111/j.1365-2486.2006.01259.x
- French, H. M. (1998). An appraisal of cryostratigraphy in north-west Arctic Canada. Permafrost and Periglacial Processes, 9(4), 297–312. https://doi.org/10.1002/(SICI)1099-1530(199810/12)9:4<297::AID-PPP296>3.0.CO;2-B
- Fritz, M., Vonk, J. E., & Lantuit, H. (2017). Collapsing Arctic coastlines. Nature Clim. Change, 7, 6–7. https://doi.org/10.1038/ nclimate3188
- Gustafsson, Ö., Widerlund, A., Andersson, P. S., Ingri, J., Roos, P., & Ledin, A. (2000). Colloid dynamics and transport of major elements through a boreal river—Brackish bay mixing zone. *Marine Chemistry*, 71(1–2), 1–21. https://doi.org/10.1016/S0304-4203(00)00035-9
- Hugelius, G., Strauss, J., Zubrzycki, S., Harden, J. W., Schuur, E. A. G., Ping, C. L., et al. (2014). Estimated stocks of circumpolar permafrost carbon with quantified uncertainty ranges and identified data gaps. *Biogeosciences*, 11(23), 6573–6593. https://doi.org/10.5194/bg-11-6573-2014
- Irrgang, A. M., Lantuit, H., Gordon, R. R., Piskor, A., & Manson, G. K. (2019). Impacts of past and future coastal changes on the Yukon coast —Threats for cultural sites, infrastructure, and travel routes. Arctic Science, 20(February), 1–20. https://doi.org/10.1139/as-2017-0041
- Jones, B. M., Farquharson, L. M., Baughman, C. A., Buzard, R. M., Arp, C. D., Grosse, G., et al. (2018). A decade of remotely sensed observations highlight complex processes linked to coastal permafrost bluff erosion in the Arctic A decade of remotely sensed observations highlight complex processes linked to coastal permafrost bluff eros. *Environmental Research Letters*, 13(11), 115001. https://doi.org/ 10.1088/1748-9326/aae471
- Jorgenson, M. T., Kanevskiy, M., Shur, Y., Moskalenko, N., Brown, D. R. N., Wickland, K., et al. (2015). Role of ground ice dynamics and ecological feedbacks in recent ice wedge degradation and stabilization. *Geophysical Research Letters Earth Surface*, 120(11), 2280–2297. https://doi.org/10.1002/2015JF003602
- Kanevskiy, M., Shur, Y., Jorgenson, M. T., Ping, C. L., Michaelson, G. J., Fortier, D., et al. (2013). Ground ice in the upper permafrost of the Beaufort Sea coast of Alaska. Cold Regions Science and Technology, 85, 56–70. https://doi.org/10.1016/j.coldregions.2012.08.002
- Karlsson, E. S., Charkin, A., Dudarev, O., Semiletov, I., Vonk, J. E., Sánchez-García, L., & Andersson, A. (2011). Carbon isotopes and lipid biomarker investigation of sources, transport and degradation of terrestrial organic matter in the Buor-Khaya Bay, SE Laptev Sea. *Biogeosciences*, 8, 1865–1879. https://doi.org/10.5194/bg-8-1865-2011
- Knoblauch, C., Beer, C., Sosnin, A., Wagner, D., & Pfeiffer, E. M. (2013). Predicting long-term carbon mineralization and trace gas production from thawing permafrost of Northeast Siberia. *Global Change Biology*, 19(4), 1160–1172. https://doi.org/10.1111/gcb.12116
- Koven, C. D., Schuur, E. A. G., Schädel, C., Bohn, T. J., Burke, E. J., Chen, G., et al. (2015). A simplified, data-constrained approach to estimate the permafrost carbon–climate feedback. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, 373(2054), 20140423. https://doi.org/10.1098/rsta.2014.0423
- Lantuit, H., Overduin, P. P., Couture, N., Wetterich, S., Aré, F., Atkinson, D., et al. (2012). The Arctic coastal dynamics database: A new classification scheme and statistics on Arctic permafrost coastlines. *Estuaries and Coasts*, 35(2), 383–400. https://doi.org/10.1007/s12237-010-9362-6
- Lantuit, H., & Pollard, W. H. (2005). Temporal stereophotogrammetric analysis of retrogressive thaw slumps on Herschel Island, Yukon Territory. Natural Hazards and Earth System Science, 5(3), 413–423. https://doi.org/10.5194/nhess-5-413-2005
- Lee, H., Schuur, E. A., Inglett, K. S., Lavoie, M., & Chanton, J. P. (2012). The rate of permafrost carbon release under aerobic and anaerobic conditions and its potential effects on climate. *Global Change Biology*, *18*(2), 515–527. https://doi.org/10.1111/j.1365-2486.2011.02519.x Lee, H., Swenson, S. C., Slater, A. G., & Lawrence, D. M. (2014). Effects of excess ground ice on projections of permafrost in a warming
- climate. Environmental Research Letters, 9(12), 124006. https://doi.org/10.1088/1748-9326/9/12/124006
- Letscher, R. T., Hansell, D. A., & Kadko, D. (2011). Rapid removal of terrigenous dissolved organic carbon over the Eurasian shelves of the Arctic Ocean. *Marine Chemistry*, 123(1-4), 78–87. https://doi.org/10.1016/j.marchem.2010.10.002
- Macdonald, R. W., Kuzyk, Z. Z. A., & Johanessen, S. C. (2015). The vulnerability of Arctic shelf sediments to climate change. Environmental Reviews, 23(4), 461–479. https://doi.org/10.1139/er-2015-0040
- McGuire, A., Anderson, L., Christensen, T. R., Dallimore, S., Guo, L., Hayes, D., et al. (2009). Sensitivity of the carbon cycle in the Arctic to climate change. *Ecological Monographs*, 79(4), 523–555. https://doi.org/10.1890/08-2025.1
- Olefeldt, D., Goswami, S., Grosse, G., Hayes, D., Hugelius, G., Kuhry, P., et al. (2016). Circumpolar distribution and carbon storage of thermokarst landscapes. *Nature Communications*, 7, 13043. https://doi.org/10.1038/ncomms13043
- Overeem, I., Anderson, R. S., Wobus, C. W., Clow, G. D., Urban, F. E., & Matell, N. (2011). Sea ice loss enhances wave action at the Arctic coast. *Geophysical Research Letters*, 38, L17503. https://doi.org/10.1029/2011GL048681
- Pollard, W. (1990). The nature and origin of ground ice in the Herschel Island area, Yukon Territory. *Proceedings of the Fifth Canadian Conference on Permafrost*, (1953), 23–30. Retrieved from http://pubs.aina.ucalgary.ca/cpc/CPC5-23.pdf
- Ramage, J. L., Irrgang, A. M., Herzschuh, U., Morgenstern, A., Couture, N., & Lantuit, H. (2017). Terrain controls on the occurrence of coastal retrogressive thaw slumps along the Yukon Coast, Canada. *Journal of Geophysical Research: Earth Surface*, 122, 1619–1634. https://doi.org/10.1002/2017JF004231
- Rasmussen, H., & Jorgensen, B. B. (1992). Microelectrode studies of seasonal oxygen uptake in a coastal sediment: Role of molecular diffusion. Marine Ecology Progress Series, 81(3), 289–303. https://doi.org/10.3354/meps081289
- Raymond, P. A., McClelland, J. W., Holmes, R. M., Zhulidov, A. V., Mull, K., Peterson, B. J., et al. (2007). Flux and age of dissolved organic carbon exported to the Arctic Ocean: A carbon isotopic study of the five largest arctic rivers. *Global Biogeochemical Cycles*, 21, GB4011. https://doi.org/10.1029/2007GB002934
- Ruz, M. H., Héquette, A., & Hill, P. R. (1992). A model of coastal evolution in a transgressed thermokarst topography, Canadian Beaufort Sea. Marine Geology, 106(3–4), 251–278. https://doi.org/10.1016/0025-3227(92)90133-3
- Schädel, C., Schuur, E. A. G., Bracho, R., Elberling, B., Knoblauch, C., Lee, H., et al. (2014). Circumpolar assessment of permafrost C quality and its vulnerability over time using long-term incubation data. *Global Change Biology*, 20(2), 641–652. https://doi.org/10.1111/ gcb.12417
- Schirrmeister, L., Kunitsky, V., Grosse, G., Wetterich, S., Meyer, H., Schwamborn, G., et al. (2011). Sedimentary characteristics and origin of the Late Pleistocene ice complex on north-east Siberian Arctic coastal lowlands and islands—A review. *Quaternary International*, 241, 3–25. https://doi.org/10.1016/j.quaint.2010.04.004
- Schreiner, K. M., Bianchi, T. S., Eglinton, T. I., Allison, M. A., & Hanna, A. J. M. (2013). Sources of terrigenous inputs to surface sediments of the Colville River Delta and Simpson's Lagoon, Beaufort Sea, Alaska. *Journal of Geophysical Research: Biogeosciences*, 118, 808–824. https://doi.org/10.1002/jgrg.20065

- Schuur, E. A. G., Bockheim, J., Canadell, J. P., Euskirchen, E., Field, C. B., Goryachkin, S. V., et al. (2008). Vulnerability of permafrost carbon to climate change: Implications for the global carbon cycle. *BioScience*, 58(8), 701–714. https://doi.org/10.1641/B580807 Schuur, E. A. G., McGuire, A. D., Grosse, G., Harden, J. W., Hayes, D. J., Hugelius, G., et al. (2015). Climate change and the permafrost
- Schult, E. A. G., McGuite, A. D., Grosse, G., Haldell, J. W., Hayes, D. J., Hugends, G., et al. (2013). Climate change and the permanosis carbon feedback. *Nature*, 520(January 2016), 171–179. https://doi.org/10.1038/nature14338
 - Semiletov, I. P., Shakhova, N. E., Pipko, I. I., Pugach, S. P., Charkin, A. N., Dudarev, O. V., et al. (2013). Space-time dynamics of carbon and environmental parameters related to carbon dioxide emissions in the Buor-Khaya Bay and adjacent part of the Laptev Sea. *Biogeosciences*, 10(9), 5977–5996. https://doi.org/10.5194/bg-10-5977-2013
- Stein, R., & Macdonald, R. W. (2004). The Arctic organic carbon cycle: Present and past. Berlin, Heidelberg, New York: Springer.
- Strauss, J., Schirrmeister, L., Mangelsdorf, K., Eichhorn, L., Wetterich, S., & Herzschuh, U. (2015). Organic-matter quality of deep permafrost carbon—A study from Arctic Siberia. *Biogeosciences*, 12(7), 2227–2245. https://doi.org/10.5194/bg-12-2227-2015
- Tanski, G., Lantuit, H., Ruttor, S., Knoblauch, C., Radosavljevic, B., Strauss, J., et al. (2017). Transformation of terrestrial organic matter along thermokarst-affected permafrost coasts in the Arctic. Science of the Total Environment, 581–582, 434–447. https://doi.org/10.1016/j. scitotenv.2016.12.152
- Tremblay, J. É., Bélanger, S., Barber, D. G., Asplin, M., Martin, J., Darnis, G., et al. (2011). Climate forcing multiplies biological productivity in the coastal Arctic Ocean. *Geophysical Research Letters*, *38*, L18604. https://doi.org/10.1029/2011GL048825
- Turetsky, M. R., Abbott, B. W., Jones, M. C., Anthony, K. W., Olefeldt, D., Schuur, E. A. G., et al. (2019). Permafrost collapse is accelerating carbon release. *Nature*, 569, 32–34. https://doi.org/10.1038/d41586-019-01313-4
- Von Deimling, T. S., Grosse, G., Strauss, J., Schirrmeister, L., Morgenstern, A., & Schaphoff, S. (2015). Observation-based modelling of permafrost carbon fluxes with accounting for deep carbon deposits and thermokarst activity, (2006). *Biogeosciences*, 12(11), 3469–3488. https://doi.org/10.5194/bg-12-3469-2015
- Vonk, J. E., Sánchez-García, L., van Dongen, B. E., Alling, V., Kosmach, D., Charkin, A., et al. (2012). Activation of old carbon by erosion of coastal and subsea permafrost in Arctic Siberia. *Nature*, 489(7414), 137–140. https://doi.org/10.1038/nature11392
- Vonk, J. E., Semiletov, I. P., Dudarev, O. V., Eglinton, T. I., Andersson, A., Shakhova, N., et al. (2014). Preferential burial of permafrostderived organic carbon in Siberian-Arctic shelf waters. *Journal of Geophysical Research: Oceans*, 119, 8410–8421. https://doi.org/10.1002/ 2014JC010261
- Vonk, J. E., Tank, S. E., Bowden, W. B., Laurion, I., Vincent, W. F., Alekseychik, P., et al. (2015). Reviews and syntheses: Effects of permafrost thaw on Arctic aquatic ecosystems. *Biogeosciences*, 12(23), 7129–7167. https://doi.org/10.5194/bg-12-7129-2015
- Vonk, J. E., & Gustafsson, Ö. (2013). Permafrost-carbon complexities. Nature Geoscience, 6(9), 675–676. https://doi.org/10.1038/ngeo1937
 Vonk, J. E., Mann, P. J., Davydov, S., Davydova, A., Spencer, R. G. M., Schade, J., et al. (2013). High biolability of ancient permafrost carbon upon thaw. Geophysical Research Letters, 40, 2689–2693. https://doi.org/10.1002/grl.50348
- Wagner, D., Pfeiffer, E. M., & Bock, E. (1999). Methane production in aerated marshland and model soils: Effects of microflora and soil texture. Soil Biology and Biochemistry, 31(7), 999–1006. https://doi.org/10.1016/S0038-0717(99)00011-5
- Walter, K. M., Edwards, M. E., Grosse, G., Zimov, S. A., & Chapin, F. S. (2007). Thermokarst lakes as a source of atmospheric CH₄ during the Last Deglaciation. *Science*, 318(633), 633–636. https://doi.org/10.1126/science.1142924
- Ward, C. P., & Cory, R. M. (2016). Complete and partial photo-oxidation of dissolved organic matter draining permafrost soils. Environmental Science & Technology, 50(7), 3545–3553. https://doi.org/10.1021/acs.est.5b05354
- Ward, C. P., Nalven, S. G., Crump, B. C., Kling, G. W., & Cory, R. M. (2017). Photochemical alteration of organic carbon draining permafrost soils shifts microbial metabolic pathways and stimulates respiration. *Nature Communications*, 8(1), 1–7. https://doi.org/10.1038/s41467-017-00759-2
- Wegner, C., Bennett, K. E., de Vernal, A., Forwick, M., Fritz, M., Heikkila, M., et al. (2015). Variability in transport of terrigenous material on the shelves and the deep Arctic Ocean during the Holocene. *Polar Research*, *34*, 1–19. https://doi.org/10.3402/polar.v34.24964